Title: Moringa Oleifera down-regulates nuclear factor-kappaB and increases cytotoxic effect of chemotherapy in pancreatic cancer cells

Authors:

Liron Berkovich (liron.berkovich@gmail.com)
Gideon Earon (drearon1@gmail.com)
Ilan Ron (prof.ilan.ron@gmail.com)
Adam Rimmon (arimmon@yahoo.com)
Akiva Vexler (akivav@tlvmc.gov.il)
Shahar Lev-Ari (Shahar1@tasmc.health.gov.il)

Version: 2 Date: 28 May 2013

Author's response to reviews: see over
To: Dr. Tom Rowles

Executive Editor, BMC Complementary and Alternative Medicine

Dear Dr. Rowles,

RE: "Moringa Oleifera down-regulates nuclear factor-kappaB and increases cytotoxic effect of chemotherapy in pancreatic cancer cells"

Thank you for allowing us to revise and re-submit our paper. Enclosed please find our revised manuscript, and here, a point by point letter addressing all the constructive comments of the reviewers.

Yours sincerely,

Dr. Liron Berkovich
Laboratory of Herbal Medicine and Cancer Research,
Sourasky Medical Center,
Tel-Aviv, Israel.